Author's response to reviews

Title: Long-term Outcome after Bilateral Lung Transplantation: A Low-Volume Center Experience

Authors:

Hsao-Hsun Hsu (ntuhsu@gmail.com)
Shu-Chien Huang (dtsurg99@yahoo.com.tw)
Shuenn-Wen Kuo (shuenn@ntuh.gov.tw)
Pei-Ming Huang (e370089@gmail.com)
Jin-Shing Chen (chenjs@ntu.edu.tw)
Jang-Ming Lee (jangminglee@gmail.com)
Sung-Ching Pan (scpbpan@gmail.com)
Wen-Je Ko (kowj@ntu.edu.tw)

Version: 2 Date: 18 March 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

We are very pleased to re-submit our manuscript entitled “Long-term Outcome after Bilateral Lung Transplantation – A Low-Volume Center Experience” submitted to *Pulmonary Medicine* for publication in the category of Research articles. We have already revised our manuscript according to your previous email and described the changes made point-by-point as following:

1. **Title Page**: Please include the email addresses of all authors.

   **Reply**: The email addresses of all authors have been included in the title page.

   (Page 2)

2. **Conclusions**: Please provide Conclusion after Discussion.

   **Reply**: We have provided the conclusion in the manuscript. (Page 15)

3. **Line Numbering**: Please revise your manuscript to include line and page numbers.

   **Reply**: The line and page numbers have been included in the manuscript.

4. **Ethics**: Please provide the name of Institutional Review Board who approved the study.

   **Reply**: The name of Institutional Review Board has been included in the
Sincerely Yours,

Hsao-Hsun Hsu, MD, PhD

Department of Surgery,

National Taiwan University Hospital,

No. 7, Chung-Shan South Road, Taipei, Taiwan

Tel: 886-2-23123456 ext. 65070

Fax: 886-2-2341-2542

E-mail: ntuhsu@gmail.com